Author's response to reviews

Title: Rare and massive odontogenic parakeratotic cyst treated by endoscopic sinus surgery: a case report

Authors:

Dario Marcotullio (dario.marcotullio@uniroma1.it)
Giannicola Iannella (giannicolaiainnella@hotmail.it)
Melissa Zelli (vibrissa82@hotmail.com)
Caterina Marinelli (jaja1978@hotmail.it)
Giuseppe Magliulo (giuseppe.magliulo@uniroma1.it)

Version: 2  Date: 2 July 2014

Author's response to reviews: see over
Dear editor we have carried out the required changes:

1) Q1: Please ensure that the Case presentation in the Abstract section includes brief details of what the patient(s) presented with, including the patient's age, sex and ethnic background. If the Case reported is of multiple patients, please enumerate them accordingly.

A1: We included in the Case presentation section of the Abstract brief details of what the patient presented including also age, sex and ethnic background.

2) Q1: Please include the Ethnicity of the patient in the Abstract and Case presentation sections.

A1: We included the Ethnicity of the patient in the case presentation sections.

3) Q1: Please include an acknowledgement section at the end of the manuscript before the reference list. Please acknowledge anyone who contributed towards the study by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for all authors. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements.

A1: We included the acknowledgement section at the end of the manuscript before the reference list.

Best regards

Giuseppe Magliulo